Author's response to reviews

Title: Factors associated with ultrasound-aided detection of suboptimal fetal growth in a malaria-endemic area in Papua New Guinea

Authors:

Holger W Unger (hwunger@doctors.org.uk)
Maria Ome-Kaius (marai.kaius@gmail.com)
Stephan Karl (karl@wehi.edu.au)
Dupain Singirok (holiveoil@gmail.com)
Peter Siba (pmaxsiba@gmail.com)
Jane Walker (Jane.Walker@nhslothian.scot.nhs.uk)
Regina A Wangnapi (wangnapi.regina@yahoo.com)
Ivo Mueller (mueller@wehi.edu.au)
Stephen J Rogerson (sroger@unimelb.edu.au)

Version: 4 Date: 8 March 2015

Author's response to reviews: see over
Dr Ndola Prata,
Editorial Board,
BMC Pregnancy and Childbirth

Melbourne, 9 March 2015

Dear Dr Prata,

Factors associated with ultrasound-aided detection of suboptimal fetal growth in a malaria-endemic area in Papua New Guinea

Thank you for your email dated March 2nd 2015.

We have revised the manuscript to include a Conclusions section, as requested (Lines 332-337).

Yours faithfully,

Prof Stephen Rogerson on behalf of the authors